Author's response to reviews

Title: Study of the protective effect on intestinal mucosa of the hydrosoluble fiber Plantago ovata husk

Authors:

Ana M Sahagun (amsahp@unileon.es)
Jose Vaquera (jvago@unileon.es)
Juan J Garcia (jigary@unileon.es)
Angela P Calle (apcalp@unileon.es)
Maria Jose Diez (mjdiel@unileon.es)
Nelida Fernandez (nelida.fernandez@unileon.es)
Hugo O Portilla (hoporp@unileon.es)
Juan F Loro (juanfrancisco.loro@ulpgc.es)
Matilde Sierra (msiev@unileon.es)

Version: 2 Date: 9 December 2014

Author's response to reviews:

Page and line numbers have been included